Author's response to reviews

Title: Inflammation gene variants and susceptibility to albuminuria in the U.S. population: analysis in the Third National Health and Nutrition Examination Survey (NHANES III), 1991-1994

Authors:

Renee M Ned (RNed@cdc.gov)
Ajay Yesupriya (AYesupriya@cdc.gov)
Giuseppina Imperatore (GImperatore@cdc.gov)
Diane T Smelser (dtsmelser@geisinger.edu)
Ramal Moonesinghe (RMoonesinghe@cdc.gov)
Man-huei Chang (MChang@cdc.gov)
Nicole F Dowling (NDowling@cdc.gov)

Version: 5 Date: 4 November 2010

Author's response to reviews:

Minor revisions to spacing between words/numbers and symbols (e.g., after = signs and between numbers and units). I also expanded a column in Table 3 to accommodate the long text. I also did minor wording changes in the Abstract and Acknowledgements.